Author’s response to reviews

Title: Medication Safety Curriculum: Enhancing Skills and Changing Behaviors

Authors:

  Kelly Karpa (kjd136@psu.edu)
  Lindsay Hom (lindsaylea9@gmail.com)
  Paul Huffman (huffmanpaulg@gmail.com)
  Erik Lehman (elehman@phs.psu.edu)
  Paul Haidet (phaidet@hmc.psu.edu)
  Shou Ling Leong (sleong@hmc.psu.edu)

Version: 2
Date: 19 September 2014

Author’s response to reviews:

Adding page numbers and line numbers to manuscript, per editorial request.